Author’s response to reviews

Title: Chronic kidney disease and HIV in the era of antiretroviral treatment: findings from a 10-year cohort study in a West African setting

Authors:

Firmin Kaboré (nongodo@yahoo.fr)
Armel Poda (armelpoda@yahoo.fr)
Jacques Zoungrana (zojacques@yahoo.fr)
Ollo Da (olobouff@yahoo.fr)
Laura Ciaffi (lauraciaffi2002@yahoo.fr)
Aoua Semdé (semdaoua@yahoo.fr)
Issouf Yaméogo (dryamis@yahoo.fr)
Adrien Sawadogo (sawadogoadrien715@gmail.com)
Eric Delaporte (eric.delaporte@ird.fr)
Nicolas Meda (nicolas.meda@gmail.com)
Sophie Limou (sophie.limou@univ-nantes.fr)
Amandine Cournil (amandine.cournil@ird.fr)

Version: 1 Date: 22 Nov 2018

Author’s response to reviews:

1. Ethics approval and consent to participate

   a) Ethics Committee aprobation is obtained for this study.

   b) Administrative permission required/granted or order to access the patient database is clarified.

2. Figure legends

   The figure legends embedded within the main text of the manuscript are removed
3. Formatting

The section headings of the manuscript are renamed and the main text include: Background, Methods, Results, Conclusions, Discussion—5 sections.

4. The revised manuscript does not contain tracked changes or highlighting.